Author’s response to reviews

Title: Frail-VIG index: a concise frailty evaluation tool for rapid geriatric assessment. Results of a 24-months prospective follow-up

Authors:
Jordi Amblàs-Novellas (jordiamblas@gmail.com)
Martori Joan Carles (martori@uvic.cat)
Joan Espaulella (jespaulella@hsc.chv.cat)
Ramon Oller (ramon.oller@uvic.cat)
Núria Molist-Brunet (nmolist@gmail.com)
Marco Inzitari (minzitari@perevirgili.cat)
Roman Romero-Ortuno (roman.romero-ortuno@nhs.net)

Version: 1 Date: 17 Oct 2017

Author’s response to reviews:

Dear Editor,

We would like to thank you for the possibility of adapting the manuscript BGTC-D-17-00534 "Frail-VIG index: design and validation of a new, concise frailty evaluation tool for rapid geriatric assessment. Results of a 24-months prospective follow-up". We have revised it according your suggestions, comments and recommendations, which we really appreciate.

We hope the amendments done on the manuscript have improved and increased its quality so as to accomplish the criteria for publication. That is why we are resending it to your kind consideration.

We look forward to receiving your response at your earliest convenience. We will be pleased to introduce further modifications to merit the publication of the manuscript in your journal. Thanks very much.